Author's response to reviews

Title: Is Lamarckian Evolution Relevant to Medicine?

Authors:

Adam E Handel (ahandel@doctors.org.uk)
Sreeram V Ramagopalan (sreeramr@well.ox.ac.uk)

Version: 2 Date: 24 February 2010

Author's response to reviews:

Dear sir or madam,

We would be very grateful if you would consider our manuscript for publication in BMC Medical Genetics. The role of epigenetics in Lamarkian evolution is one that has too frequently been ignored in the literature of medical genetics. Our commentary article attempts to highlight this potentially very important area.

Best wishes,
Adam & Sreeram

Dr. Adam E Handel & Dr. Sreeram V Ramagopalan,
Wellcome Trust Centre for Human Genetics,
University of Oxford,
Oxford,
UK